Author's response to reviews

Title: Randomised controlled trial of a secondary prevention program for myocardial infarction patients ('ProActive Heart'): study protocol

Authors:

Anna L Hawkes (annahawkes@cancerqld.org.au)
John Atherton (john_atherton@health.qld.gov.au)
Barr Taylor (btaylor@stanford.edu)
Paul Scuffham (p.scuffham@griffith.edu.au)
Kathy Eadie (kathyeadie@cancerqld.org.au)
Brian Oldenburg (brian.oldenburg@med.moansh.edu.au)

Version: 2 Date: 7 April 2009

Author's response to reviews: see over
6\textsuperscript{th} April 2009

Editor-in-Chief
Dr Melissa Norton, MD
BMC-Cardiovascular Disorders

Dear Dr Norton

Please find attached a revised document for manuscript ID:1075954011259100 including reference to ethics approval as requested. Thank you for considering our manuscript for publication and I look forward to your response.

Yours sincerely

Anna L Hawkes
Senior Research Fellow
Viertel Centre for Research in Cancer Control
Cancer Council Queensland
553 Gregory Terrace, Fortitude Valley QLD 4006
AnnaHawkes@cancerqld.org.au
Ph: +61 7 3258 2305
Fax: +61 7 3258 2310